UNDERSTANDING AND EVALUATING SUSTAINABLE URBAN

TRENDS: CASE STUDIES FROM CHINA AND JAPAN

(SUMMARY)

By

Xiaolong ZOU

Dissertation Submitted to Graduate School of Asia Pacific Studies, Ritsumeikan Asia Pacific University

in partial fulfillment of the requirement for the degree of Doctor of Philosophy in Asia Pacific Studies

Beppu, Japan, 2016

© Xiaolong Zou

DECLARATION

Due to certain copyright issues, partial contents of chapters that are under the consideration for publication or peer evaluation have not been made available. Please contact the author directly if you wish to obtain further information regarding this dissertation.

ABSTRACT

The advent of the Brundtland's Report *Our Common Future* in 1987 and the successful launching of the Rio Earth Summit in 1992 has built international consensus on "sustainability" as a new paradigm for development. The development of "eco-cities" has become an international phenomenon for the creation of more sustainable urban areas. Subsequently, the negotiation of the Kyoto Protocol in 1997 supported another global wave of "low-carbon cities" development. Additionally, since the early 2000s, the development of information and communication technologies has become the impetus for an innovation-oriented sustainable urban trend known as the "smart cities". However, despite the enthusiastic advancement of these new urban models worldwide, there is still a lack of consensus regarding systematic approaches or methods for the standardization and evaluation of these trends.

This thesis aims to investigate and examine three global trends of sustainable cities with case studies from China and Japan, in both quantitative and qualitative perspectives, to understand their defining features and components. Furthermore, this thesis intends to propose and develop a methodical approach for the evaluation of these urban development models to have flexibility in relation to local inputs, and applicability to other similar urban initiatives or projects.

For the "eco-cities", this thesis reviews studies regarding concepts, frameworks and indicator systems, A large amount of literature on the selection of indicators under a singular framework in China is observed rather than having a quantity comparison from a broader scope. To obtain a quantitative sense of how effective China's eco-cities are compared to other best practice in the international arena, two cases from China and Japan have been selected to examine their indicator values under the national eco-city framework of China. Gaps between economy related indicator values are identified, suggesting lower average economic values and energy efficiencies of Chinese eco-cities. Targets concerning the waste sector are also lower for China than in Japan. The environmental indicator values show lower levels than in the other two cases as well, while social indicators entail a specific methodological approach for measurements in China. Suggestions are made in the discussion section based on the outcomes of the aforementioned comparisons, to provide a reference for the future development of other eco-cities.

The ensuing study on low-carbon cities employs a qualitative view of these policies such as "garden city" to "low-carbon city" to determine the how the environment-related urban environmental policy developed during different periods in China. Case studies of leading low-carbon cities are examined and analyzed to obtain insights regarding their urban environmental policies as well as the implications of their successes and limitations. The major findings indicate that government policy and financial support played a significant role in transforming the industrialized city of Kitakyushu into a center of low-carbon sustainable practices in Japan's case. Local autonomy and flexibility in policymaking and civic participation profoundly contributed to the successful switch to renewable energy. These experiences could serve as useful references for China's low-carbon city development from different perspectives.

Next, the literature regarding smart city phenomena is thoroughly reviewed. Despite a lack of universal consensus, there seems to have been two major streams of SC concepts with overarching strategies for comprehensive SC development, with specific focuses on utilizing information and communication technologies to improve the quality of life. Key features and components of smart cities are then summarized, consolidated into a proposed framework consisting of two main objectives, six domains, and two means for implementation. Furthermore, a customized smart city index for the City of Kitakyushu in Japan is proposed as a case example for the application of the proposed framework. The outcomes of this section provide new approaches for understanding smart city concepts and evaluating the on-going smart cities in Japan and potentially in other countries.

As a continuation of the previous section, a further refined selection of indicators from the proposed smart city index based on stakeholder inputs from Kitakyushu City is conducted. These indicators are then weighted by expert opinion surveys using the analytical hierarchy process (AHP) method. This weighted smart city index can be useful for prioritization of policy implementation or selection of key performance indicators (KPIs). More importantly, this integrated approach consisting of three main steps from conceptual understanding to index development and indicator weighting is found to be customizable and potentially applicable to other urban development models in different local settings. This finding would contribute to a more insightful understanding of sustainable urban projects and their evaluations for policy makers, urban planners and city managers.

The findings and outcomes of this dissertation contribute to the existing literature on urban sustainability with elaborated studies on "Eco-city", "Low-carbon City" and "Smart City" in terms of comprehension and evaluation. The conceptualized integrated method for urban development evaluation can offer practical references for policy makers, and urban managers, as well as to academia for further research.

ACKNOWLEDGEMENT

Confucius once said, "At fifteen, I set my heart upon learning. At thirty, I planted my feet firm upon the ground." In my case, I didn't know just how little I knew at fifteen. At thirty, I now know that I know little, especially upon the completion of this dissertation. Pursuing a PhD is perhaps the most challenging endeavor I have undertaken, and it would never have been remotely possible without the help, support and encouragement from these people in my life.

First and foremost, I would like to express my utmost gratitude and appreciation to my supervisor and mentor – Professor Yan Li. I was once in the crossroads of my life hesitating whether to go forth with the pursuit of doctoral study, when Prof. Li graciously took me on. From her, I have witnessed and learned the real qualities of a scholar and an educator, who is driven by genuine interest in the creation of knowledge and selfless dedication in mentoring students. Her meticulous attitude towards research and scientific study has become my constant reminder of how research should be conducted. Without her guidance, I couldn't have achieved the same level of academic outcomes as of today.

Additionally, I also would like to thank Prof. Yasushi Suzuki, Prof. Shunzo Tsukada, Prof. Hiromi Yamashita, Prof. Xuepeng Qian, Prof. Kozo Otsuka, Prof. Faezeh Mahichi and Prof. Yuichi Kotake. I had the pleasure of either sitting in their classes, lectures and seminars, or assisting them in class, workshops or research projects, where I was deeply impressed by their expertise. I'll be forever grateful for the help, suggestions, and encouragement they granted to me over the years. I also want to express my sincere gratitude to Prof. Francisco Fellizar and Prof. Robert Salazar, Prof. Malcolm Cooper and Prof. Chaolin Gu, whose advices and comments greatly helped me to improve my dissertation.

No one survives a life without friends, let alone a life of PhD pursuit. I am lucky to have the most reliable and available "comrades" through the journey thus far. Mr. Liguo Wang (the future Dr. Wang) is my most devoted listener, counselor, and confidant, without whom, I'd probably have more sorrowful days than pleasant ones during my campus life.

Ms. Xiaoai Xia is always the most dependable and truthful friend to me, though sometimes, her frankness could be a blow to my self-esteem.

My master's classmate, great friend and colleague, Ms. Christine Meister is always there to help and show support for my work, and I know our future cooperation will yield more fruits. My awesome drinking buddy and master's classmate, Mr. Tsuyoshi Nakahara and his lovely family for every time he would take me to some great sashimi restaurant in town and we always had the most meaningful conversations. My colleague and the future Dr. Tawhid Monzur, who is probably the most diligent student I've ever met in person, with whom I spent numerous lunches, is a true inspiration for me in academic life.

Last but not least, I'd like to thank my partner in life Miss Zhanyan Liu, without whom, I'd definitely be being finishing this dissertation earlier, but would undoubtedly be less happy and fulfilled. I am eternally beholden and grateful to my mom and dad, uncle Zhao and those relatives who believed in me. Without their sacrifice and support, I would never have become the man I am today.

DEDICATION

To my most beloved father, who passed away on July 11, 2016

TABLE OF CONTENTS

ABSTRACT	II
ACKNOWLEDGEMENT	IV
DEDICATION	VI
TABLE OF CONTENTS	VII
LIST OF FIGURES	X
LIST OF TABLES	XI
ACRONYMS	XIII
1 INTRODUCTION	1
1.1 BACKGROUND OF RESEARCH	1
1.2 THESIS OBJECTIVES AND RESEARCH QUESTIONS	
1.3 Adopted Methodologies	5
1.4 STRUCTURE OF THE DISSERTATION	6
2 REVIEW OF LITERATURE	9
2.1 URBAN DEVELOPMENT TRENDS TOWARDS SUSTAINABILITY	9
2.1.1 Historical Background	9
2.1.2 Sustainable Urban Categories and Trends	11
2.1.3 From Garden City to Eco-village	
2.1.4 Eco-cities	14
2.1.5 Low-carbon Cities	19
2.1.6 Smart Cities	21
2.2 PROMINENT URBAN THEORIES	
2.3 URBAN SUSTAINABILITY EVALUATION	
3 ECO-CITY DEVELOPMENT IN CHINA: INTERNATIONAL PERS	PECTIVE
AND COMPARAISON	
3.1 Chapter Introduction	
3.1.1 Eco-city origin and concepts	
3.1.2 Definitions of eco-city	

	3.2 Di	EVELOPMENT OF THE ECO-CITY	
	3.2.1	Eco-city development	
	3.3 Ec	CO-CITIES DEVELOPMENT IN CHINA	
	3.3.1	Eco-city framework in China	
	3.3.2	The Eco-cities indicator systems in China	
	3.4 Co	OMPARISON FOR CHINA'S ECO-CITIES WITH JAPAN'S CASES	
	3.4.1	Selection of case study cities	49
	3.4.2	Data collections	49
	3.4.3	Comparison criteria and applied method	50
	3.5 Ec	CO-CITY COMPARISON BETWEEN CHINA AND JAPAN	50
	3.5.1	Economic aspect	50
	3.5.2	Environmental aspect	53
	3.5.3	Social aspect	58
	3.6 CI	HAPTER CONCLUSION AND DISCUSSION	60
4]	LOW-CA	RBON CITY DEVELOPMENT IN CHINA: LESSONS AND	
RE	FERENC	CES FROM OTHER COUNTIRES	63
	4.1 Ci	HAPTER INTRODUCTION	64
	4.2 Di	EVELOPMENT OF CHINA'S MAJOR NATIONAL URBAN POLICIES	
	4.2.1	International background	66
	4.2.2	China's low-carbon urban policy developments	66
	4.3 CI	HINA'S LOW-CARBON ERA	68
	4.4 Le	ESSONS AND IMPLICATIONS FROM OTHER COUNTRIES	72
	4.4.1	Case Study of Japan: Kitakyushu Eco-Town Project	72
	4.4.2	Case Study of Germany: Rhein-Hunsrück District – Renewable Ene	ergy 78
	4.5 CI	HAPTER CONCLUSION	
5 1	U NDERS	TANDING SMART-CITY DEVELOPMENTS: A NEW	
FR	AMEWO	PRK AND ITS APPLICATION IN JAPAN	83
	5.1 Ci	hapter Introduction	Q /
	5.1.1	Smart City: A Global Background	
	5.1.1 5.1.2	Smart City: A Global Background Smart city in Japan	

5.2 Smar	T CITY: THEIR ORIGIN, CONCEPT AND INDICATORS	87
5.2.1 Sr	nart city origins	87
5.2.2 Sm	nart city concepts	88
5.3 Smar	RT CITY INDICATOR SYSTEMS	94
5.4 Prop	OSED SMART CITY FRAMEWORK BASED ON THE LITERATURE REVIEW	w 98
5.5 Smar	RT CITY EVALUATION IN JAPAN: FRAMEWORK APPLICATION IN	
KITA	KYUSHU	99
5.5.1 Se	election of case study: the City of Kitakyushu	99
5.5.2 Pr	roposed Smart City Index for Kitakyushu City	102
5.6 Chap	TER CONCLUSION AND DISCUSSION	105
6 NEW EVAL	UATION APPROACH FOR SUSTAINABLE CITIES: FROM	
SMART CITY	CONCEPT TO INDICATOR WEIGHTING	107
6.1 Chap	ter Introduction	108
6.2 Weig	HTING OF INDICATORS BY AHP	109
6.3 Meth	IODOLOGY	110
6.3.1 A	Smart City Conceptual Framework	111
6.3.2 Sm	nart City Index (Indicator Selection)	114
6.4 Weig	HTING OF SMART CITY INDICATORS (USING AHP)	116
6.5 SURV	EY RESULTS AND DISCUSSION	121
6.6 Chap	TER CONCLUSION AND DISCUSSIONS	126
7 CONCLUSI	ONS	129
7.1 Снар	TER INTRODUCTION	129
7.2 Rese	arch Findings and Contributions	130
7.2.1 Re	esearch Findings	130
7.2.2 Re	esearch Contributions	132
7.3 Limit	ATIONS	133
7.4 Futu	RE RESEARCH PERSPECTIVES	135
DEEDENICES	1	125
KEFEKENCES)	137
APPENDIX		150

LIST OF FIGURES

Figure 3-1 Chronological Development of Eco-city	32
Figure 3-2 Conceptual Perspectives of Eco-City Development	33
Figure 3-3 Eco-cities framework development in China.	34
Figure 3-4 Numbers of Indicators by Major Category	42
Figure 4-1 China's Major National Sustainable Urban Policies	68
Figure 4-2 Chinese cities expressing the goals to pursue or adopt eco-city or low-carb city development goals or plans	oon 70
Figure 4-3 National Yawata Steel Works at Present.	73
Figure 4-4 "Seven Color Smoke" and "Dokai Bay" in the Past	74
Figure 4-5 Kitakyushu in its Past and Presence	76
Figure 4-6 The Components of Kitakyushu's Eco-Town Project	77
Figure 4-7 Conceptual Developments for Rhein – Hunsrück District's Urban Policies	79
Figure 4-8 Renewable Applications in Rhein – Hunsrück wind-and solar	80
Figure 5-1 A Conceptual Framework of Smart Cities	99
Figure 6-1 Integrated Approach for SC Concept framework, its SC Index and Weighti Indicators	ng of 111
Figure 6-2 Analytical frameworks for smart city concepts	112
Figure 6-3 Conceptual framework for Smart City proposed by the author	113
Figure 6-4 A General AHP Hierarchy Model	117
Figure 6-5 Customized AHP Hierarchy Structure	117

LIST OF TABLES

Table 2-1 Number of Urban Cetegories Appeared in Scopus Databus Search	. 12
Table 2-2 National Policies of Sustainable City Development in China & Japan	. 12
Table 2-3 List of Sustainbility Evaluation Categories and Indicies	25
Table 3-1 List of Entitled Cities of Different Standards by July 2011	38
Table 3-2 Major Indicator Systems in China	. 40
Table 3-3 Categories Covered in Indicator Systems	. 41
Table 3-4 Summary of Reviewed Indicator Systems	. 43
Table 3-5 Primary and Secondary Categories of Chinese Indicator Systems and International Indicator Systems	45
Table 3-6 Comparisons of Economic Indicators for 'Eco-Cities'	. 51
Table 3-7 Comparisons of Environmental Indicators for 'Eco-Cities'	. 53
Table 3-8 Selected indicators for air quality comparisons	. 56
Table 3-9 Selected indicators for water quality comparisons	. 57
Table 3-10 Comparisons of social indicators for 'eco-cities'	. 59
Table 5-1 Concepts or Definitions of Smart City Reviewed	. 90
Table 5-2 Digest of smart city indicator systems	96
Table 5-3 Selected information of Kitakyushu City	100
Table 5-4 Proposed smart city indicator system of Kitakyushu City	104
Table 6-1 Modifed SC Index for Kitakyushu City	115
Table 6-3 Weighting for Smart City Governance	121

Table 6-4 Weighting for Smart City Economy	. 122
Table 6-5 Weighting for Smart City People & Urban Living	. 123
Table 6-6 Weighting for Smart City Infrasctructure	. 124
Table 6-7 Weighting for Smart City Energy & Mobility	. 124
Table 6-8 Weighting for Smart City Environment	. 125

ACRONYMS

ACEF	ACEF
AHP	AHP
AHP	Aggregated Weights
BOD	Biochemical Oxygen Demand
CAS	Chinese Academy of Sciences
СО	Carbon Monoxide
COD	Chemical Oxygen Demand
CR	Consistency Ratio
EIU	Economist Intelligence Union
EU	European Unions
EV	Electric Vehicles
GCR	Group Consensus Ratio
GDP	Gross Domestic Product
GHG	Green House Gases
ICE	International Electronic Commission
ICT	Information and Communication Technologies
IGES	Institute for Global Environmental Strategies
ISO	International Organization for Standardization
KPI	Key Performance Indicators
MAB	Man and Biosphere
MAUT	Multi-attribute utility theory
MEP	Ministry of Environment Protection
METI	Ministry of Economy, Trade and Industry

MoC	Ministry of Construction
MHURD	Ministry of Housing, Urban-Rural Development
NAIADE	Novel Approach to Imprecise Assessment & Decision
NDRC	National Development and Reform Committee
NGO	Non-Governmental Organization
NO2	Nitrogen Dioxide
NPO	Non-for Profit Organization
OECD	Organization for Economic Co-operation and
Р	Prosperous
PM10	Particulate Matter of 10 Microns in diameter or smaller
PM2.5	Particulate Matter of 2.5 Microns in diameter or smaller
PPM	Parts Per Million
QoL	Quality Of Life
RGMM	Raw Geometric Mean Method
SC	Smart City
SEG	System Evaluation Group
SO2	Sulfur Dioxide
UN	United Nations
UNEP	United Nations Environmental Programme
UNESCO	United Nations Educational, Scientific and Cultural

1 INTRODUCTION

1.1 Background of Research

Twenty-five centuries ago, the great philosopher Aristotle defined cities as "Built Politics". In the 21st century, cities are the most complex and dynamic eco-systems globally and the centers of scientific, cultural and social innovations (Glaeser, 2011; Hall, 1998), yet still face human-lead, dynamic and shifting challenges (Mega & Pedersen, 1998). The recorded history of human civilization can also be identified as the process of globalization and urbanization (Calderoni et al., 2012).

According to the United Nations (UN), the global population reached 7.2 billion in 2013 and is expected to reach 8.1 billion by 2025, and 9.6 billion by 2050. More than half of them (53% in 2015) are living in urban areas, and the urbanization rate is expected to reach 59.9% by 2030, and 67.2% by 2050 (United Nations, 2015). Continent-wise, the highest urbanization rates are seen in North America (81%), Latin America and the Caribbean (80%), Europe (73%), and Oceania (70%) while Asia and Africa's urbanization rates are still below the world average with 47% and 40% perceptively (United Nations, 2015).

The promising prosperity of civilization that advanced alongside fast urbanization also brought devastating "side effects" for human societies such as resource scarcity, energy crisis, eco-and-environmental hazards, climate changes related disasters, slums, poverty, pandemic and the list goes on. Particularly the developing countries in Asia and Africa are experiencing the world's worst urban environmental pollutions, and this poses enormous threat to human health. The United Nations Environmental Programme (UNEP) estimated that urban air pollution causes one million premature deaths each year and costs 2% of the GDP in developed countries and 5% in developing countries (Fook & Gang, 2010, p. 2).

Long before the creation of modern days' environmental urban policy, great visionaries like Ebenezer Howard (1850-1982) had already appealed for a new way of harmonious human nature relationship – the "Garden City" (Imura, 2010). As global consensus gradually aligned on the realization that "business as usual" development mode

could no longer guarantee the effective long-term prosperity for human societies, the pursuit of a different paradigm for development has been explored continuously, particularly in the urban development context.

The subsequent mergence of "New Towns" of the UK spread around the world and shaped the modernist urban planning doctrine during the rapid urbanization after the Second World War (WWII). The Ecological Modernization (EM) was developed as a macro-theoretical model addressing the importance of sustainable development that emerged at a later stage. Other concepts like "Regenerative Development" and "Positive Development" also presented different approaches in addressing more effective ways to plan and develop cities from complex social, ecological and physical challenges in the urban sphere(Boğaçhan, 2016).

Two significant publications in the 1970s and 1980s initiated and accelerated the global consciousness on "sustainable development" or "sustainability", namely, the Club of Rome's *Limits to Growth* (Meadows et al., 1972) and the Brundtland Commission's Report *Our Common Future* (World Commission on Environment and Development, 1987). The first one, though comparatively less acknowledged than the later one, presented from an historical point of view the arguments made by Thomas Malthus in a modern context regarding population overgrowth, and the excessive burdens imposed on ecological limits of agricultural products leading to the consequently depopulation. And over one decade later, the Brundtland Report brought forth the concept of "sustainable development" which, though vague in definitive terms, heated up the conversations and dialogues in the world's forums for leaders of the globe (Holden et al., 2008).

Later in 1992, when the United Nations Conference on Environment and Development (also known as the "Earth Summit") took place in Rio de Janeiro, "sustainable development (or sustainability)" became the core principle for urban and environmental developments. After the ratification of the Kyoto Protocol (and a range of subsequent international conventions) by the majority of countries in 1997, "low-carbon" became a "new norm" for sustainable development. These international conventions and protocols have given rise to some new global trends for urban development such as "Green City", "Eco-City", and

"Low-carbon City", with the most recent urban trend under the information era – known as the "Smart City".

Post-colonial urban theory emphasizes that cities vary in shapes, sizes and forms with distinctive cultural and historical backgrounds. Therefore urban development needs to be framed into local context and contents. Though seeking universality in single sustainable urban framework or indicator set may appear dimly possible, having a certain flexible and customizable method or approach that is subject to locality would presumably benefit project developers and stakeholders.

Numerous sustainable city (or urban) projects, initiatives and programmes have been developed or being pursued on globally. They vary in geographical features, sociodemographic contexts, and implementation scales. On the one hand, various organizations, institutes and scholars have spent great efforts in developing relevant concepts, frameworks or indicator systems (or index) under the broad sustainability framework: on the other hand, there is not yet any single or universally accepted framework that applies to all the different conditions of various regions.

This thesis aims to examine the three most recent urban development trends, namely, "eco-cities", "low-carbon cities" and "smart cities" under an East Asia setting. China and Japan are selected as the two major case study areas (with other international references) to provide better understandings of these urban trends regarding concepts, frameworks and evaluation method such as indicator systems or indexes. And finally, a method or approach with high customizability from local context is proposed based on the previous steps taken. This thesis would fill in the existing research gap of lacking such method in understanding and evaluating urban trends, and shed some light on meaningful and practical methodologies for urban studies.

1.2 Thesis Objectives and Research Questions

The sustainable urban development trend or sustainable cities continuum manifest an array of different concepts, models and categories. Concepts like "Garden Cities", "Green

Cities", "Eco-Cities", "Low-carbon Cities", "Intelligent Cities", "Smart Cities" and many more can all be regarded as part of the overarching "Sustainable City" metanarrative.

The objectives of this thesis, besides answering the research questions, are to distill or propose a customizable and pragmatic method or approach for understanding the urban development trends and their evaluations. The outcomes of this thesis would provide some insightful references in translating the urban policy goals and objectives into reality with different local conditions.

This thesis revolves around two overarching research questions distilled from this era of urban sustainability seeking and discerning, namely,

1) How to understand sustainable urban development trends such as "ecocity", "low-carbon city" and "smart city" in specific local contexts;

2) How to analyze and evaluate these sustainable urban trends using a proper methodical approach with local and regional inputs.

To answer these two major research queries, a set of subsidiary questions was developed under the urban sustainable trends selected for this thesis. These urban trends can be developed into two parts according to their chronological occurrence and relevance. The first tackles the topics of "eco-city" and "low-carbon city"; the second part discusses and analyzes "smart city" and the "evaluation scheme or method". The following arrangement of questions have been embedded in each chapter to navigate the research flows of this thesis. The specific research questions are detailed in their corresponding chapters:

Regarding "Eco-cities"

- What are "eco-cities" in terms of origin, concept, frameworks and indicator systems? And what is the current status of eco-city development in China and Japan?
- Given the many studies regarding eco-cities, how exactly are China's eco-cities performing compared to other best practices on a global stage?

Regarding "Low-carbon Cities"

- What constitute "low-carbon cities" and their development status globally? And what is the current status quo of developing low-carbon cities in China with international references.
- What could be the implications or references from international examples for China's?

Regarding "Smart Cities"

- What is the current status quo of "smart cities" given their infancy stages as the newest global trend for urban development?
- Given the diversity in its interpretation, are there any common features that smart cities should incorporate or mutual framework for its comprehensions?

Regarding "Evaluation Scheme or Methods"

- What are the current evaluation systems for smart cities, and how are they evaluated?
- What could be effective methods or mechanism for smart city index or indicator systems for evaluating smart cities or other sustainable cities?

1.3 Adopted Methodologies

There are several methods adopted and applied for the four major research packages included in Chapter 3 to Chapter 6. The specific steps and descriptions are detailed in each chapter; here I generally summarize the major methodological approaches:

In Chapter 3, I first conduct systematic reviews of eco-cities regarding definitions, frameworks, indicators and related works both abroad and in China. I take a quantitative approach for comparing and analyzing China's eco-city standards with a best-practiced case study from Japan, to obtain a quantified sense of their performances for later policy analysis

and recommendations. The data used for these comparisons are from governmental records and documents, official statistics, in some cases, interviews are conducted for specific information or data that are not available through published records.

Chapter 4 starts with an in-depth review of the low-carbon city and its relative development status quo focused in China and Japan. Another Case of a German City is included as comparison. Different from the previous chapter, a qualitative approach is adopted for answering my research questions. Field trips are conducted for data collection, in-person interviews for case study analysis, based on which, a series of policy recommendations is proposed regarding the low-carbon city developments in China. Secondary data were also used in the absence of primary data for analysis.

The next two chapters (5 & 6) take an integrated approach combining both qualitative and quantitative methods. After thoroughly reviewing the literature on smart city concepts and framework, and a policy analytical pool is then applied to analyze them. Based on these results, an encompassing conceptual framework is proposed and applied with a case study in Japan. I then propose an index with carefully designed indicator identification and selection and steps, and establish a compete index for smart city evolutions. Then I conducted an experts survey for quantitatively weighting of the indicators. And finally, the approaches and steps taken were summarized into an integrated method for the evaluations of sustainable urban development models.

Literature and data are from peer-reviewed publications, and published governmental or organizational records with high credibility. Primary data for the indicator selections are collected in workshops of local stakeholders in Kitakyushu City. Indicator weighting were calculated by the excerpt survey results.

1.4 Structure of the Dissertation

The remaining parts of this dissertation are arranged and summarized as follows:

Chapter 2 reviews the major works, theories and thoughts regarding urban studies in general, and concepts and development of the studies or researches on eco-cities, low-carbon

cities, and smart-cities. This chapter offers general information regarding both theories and current practices of the topics enlisted in the thesis.

Chapter 3 investigates the urban policy frameworks and the current practices in China's major cities, with comparative introductions between the past standards of Chinese eco-cities and that of the current ones, on both national and provincial levels. Furthermore, this chapter compares the eco-city standards with the Suzhou Case in China and an international acknowledged Japanese eco-city of Kitakyushu by analyzing the key indicators from selected eco-city case studies.

Chapter 4 reviews these policies with particular focus on the "low-carbon" cities in China. Additionally, two case studies of Kitakyushu city in Japan and Rhein-Hunsrück District in Germany are examined and analyzed to obtain enlightening factors in terms of their urban policies as well as the references and implications from their successes and limitations. These experiences could offer insightful references to China's low-carbon urban developments from different perspectives.

Chapter 5 summarizes the key features and components of smart city and proposes a conclusive framework for smart cities that consists of double-objectives, six domains and two means for its realization. Furthermore, this chapter proposes a customized indicator system based on the SC framework for measuring the "smartness" of the smart cities in Japan, and includes a case study of the city of Kitakyushu. The outcome of this chapter provides some new insights to the methodological approaches adopted to assess the on-going smart city initiatives in Japan.

Chapter 6 further improves the proposed smart city conceptual framework of Chapter 5. Under this framework, I have further refined the selection of smart city indices based on the inputs from the stakeholders in the City of Kitakyushu. Revisions and modifications have been made to the proposed SC Index. The Analytical hierarchy process (AHP) is applied for the weighting of indicators by experts' survey. Finally an integrated approach is recognized as the outcome. This integrated approach is found to be highly customizable and adoptable for potential applications to other urban development models in different contexts for both framework development and index composition.

Chapter 7 summarizes the major findings of the thesis and denotes the contributions and significance of the research outcomes as well as the limitations for each conducted topic. Further discussions are conducted and the possible perspectives and directions are pointed out for future research.

2 REVIEW OF LITERATURE

This chapter¹ reviews some prominent urban theories are claimed to be the fundamental frameworks of the urban studies discipline. Major definitions and concepts, frameworks and historical developments, as well as major and influential studies are reviewed regarding a range of topics from Garden City to Eco-Village and Eco-City, from Low-carbon City to Smart City. Due to the arrangement of the thesis, additional and more specific reviews of literature are conducted in subsequent chapters.

¹Some of the contents regarding eco-villages and low-carbon cities from this chapter have been published as two book review articles in the journal *Asia Pacific World (Zou, 2015a, 2015b)*.

3 ECO-CITY DEVELOPMENT IN CHINA: INTERNATIONAL PERSPECTIVE AND COMPARISON

China's growing international dominance and influence has been witnessed and acknowledged across the world. But its annual GDP growth contributed by rapid urbanization and industrialization comes with severe environmental costs, especially in urban areas. Determined not to repeat some industrialized countries' mistake of "treatment of environment comes after the development of economy", China has set up a number of laws and regulations to safeguard the sustainable development in urban areas. Eco-city development is one of the early national attempts in curbing the derailed urbanization trajectory.

This chapter² reviews the urban policy frameworks and current practices in China's major cities, with the comparative introductions between the past standards of Chinese ecocities and that of the current, on both national and provincial levels. Furthermore, this chapter compares the existing eco-city standards with international acknowledged examples in Japan and Germany by analyzing the key indicators from selected eco-city case studies.

² Based on this chapter, a journal paper has been published as Zou, X. & Li, Y. (2014). "How 'Eco' are China's Eco-Cities: An International Perspective", *International Review for Spatial Planning and Sustainable Development*, 2(3), p18-30.

4 LOW-CARBON CITY DEVELOPMENT IN CHINA: LESSONS AND REFERENCES FROM OTHER COUNTIRES

Succeeding the global urban trend of eco-city development, the low-carbon city became the next "norm" for a new trend of urban development towards sustainability. "Low-carbon" has become the very core element of this wave. To pursue the urbanization in a sustainable and low-carbon manner, the Chinese government has strenuously enacted an array of corresponding urban policies. This chapter³ reviews these policies with particular focus on the "low-carbon" cities in China, and finds that China's major objective towards low-carbonization is to reduce CO2 emissions with proper adaptation plans. A strong focus is placed on governmental interventions that result in positive civil effects regarding carbon reduction. Additionally, two case studies from other countries are introduced to offer lessons and references for China's low-carbon city development.

³ Bases on this chapter, a paper has been published as Zou, X., & Li, Y. (2015). "Developing Tailor-Made Urban Environmental Policies for China's Low Carbon Cities - Implications from Japan and Germany". In Feng, S., Huang, W., Wang, J., Wang, M. & Zha, J. (Eds.), *Low-carbon City and Newtype Urbanization*. (pp. 273-284): Springer Berlin Heidelberg.

5 UNDERSTANDING SMART-CITY DEVELOPMENTS: A NEW FRAMEWORK AND ITS APPLICATION IN JAPAN

The Smart City (SC) concept is a new global trend for urban development, and is now gaining incremental popularity worldwide. Through an extensive literature review, it is found that despite lacking universal consensus, there have been two major streams of SC concepts with overarching strategies for comprehensive SC developments or with specific focuses on utilizing information and communication technologies (ICT) to improve the quality of life.

This chapter⁴ summarizes the key features and components of smart cities and proposes a conclusive framework for smart cities that consist of double-objectives, six domains and two means for its realization. Furthermore, this chapter proposes customized indicator system based on the SC framework for measuring the "smartness" of the smart cities in Japan based on a case study of the City of Kitakyushu (or Kitakyushu City). This chapter provides some new insights to the methodological approaches adopted to assess the on-going smart city initiatives in Japan.

⁴ Based on this chapter, a journal paper is to be published as Zou, X., & Li, Y. (2016). "Recapitulating Smart City Concepts: A Proposed Framework and its Application in Japan". *International Review for Spatial Planning and Sustainable Development*.

6 A NEW EVALUATION APPROACH FOR SUSTAINABLE CITIES: FROM SMART CITY CONCEPT TO INDICATOR WEIGHTING

In this chapter⁵, further improvement is made to the proposed smart city conceptual framework as outlined in Chapter 5. Under this framework, I have further refined the selection of indicators based on the inputs from the stakeholders in the City of Kitakyushu. Revisions and modifications have been made to the proposed index. The Analytical Hierarchy Process (AHP) is applied in the weighting of indicators by an expert survey. Finally an integrated approach is developed as the outcome.

This integrated approach is found to be highly customizable and adoptable for potential applications to other urban development models in different contexts for both framework development and index composition. The findings of this study would contribute to a more insightful understanding of the smart city and its evaluation for policy makers, academia, urban managers, and practitioners. Furthermore, this integrated approach can also be adapted to understand and evaluate sustainable cities in general with local inputs.

⁵ A journal paper based on this chapter is being prepared for journal submission.

7 CONCLUSIONS

7.1 Chapter Introduction

Human history is somehow a history of urbanization and globalization. Cities are the complex and dynamic systems where the exchanges of information, economic and social activities. With the speed of technology advancement and industrialization, development-led prosperity became a low-hanging fruit for the urbanized sphere, and contributed to the formation of more cities at even bigger scales. In the mean time, urban "illnesses" resulted from the disruptions of ecological system and ambient environment, climate changes, energy and resource crisis, various millennium challenges have devastated human societies on planet earth. A paradigm shift or change is desperately craved and needed from the long-term perspective. Starting from the early UK's urban model of "Garden Cities" in late 1890s, continuous trends of pursuing in the pursuit of sustainable cities, such as "Eco-cities", "Low-carbon Cities" and "Smart Cities" have been seen on a global scale, with incremental momentum in the past decades.

Despite the multitudinous sustainable city projects developed or being undertaken worldwide, there are still gaps in comprehending or understanding the constitutions of these sustainable urban development models. Moreover, the vacancies in systematic approaches or methods for standardizing and evaluating these sustainable movements or urban trends have made developing such projects within urban boundaries quite ambiguous and imprecise in terms of policy implementation and regional replicability. This thesis attempted to answer the overarching questions of how to understand and how to evaluate these sustainable cities of major sustainable urban trends in an East Asia setting, with selected case studies from two leading powers of China and Japan.

Amongst a number of urban categories or development models, three major sustainable urban trends were identified and selected for this thesis, namely, "eco-cities", "low-carbon cities" and "smart cities" based on both academic and country policy relevance. The first identified trend of "eco-cities" is studied in a Chinese context with a comparison of Japan's case, in terms of the concept, definition and indicator systems. The second trend of "low-carbon cities" also focused on China's situation with two case studies from Japan and Germany for comparison and reference. The third urban global trend of smart cities was framed within the Japanese background for in-depth analysis with case study of the City of Kitakyushu. And an integrated approach from concept understanding to index for evaluation and final weighting of indicators was distilled that can be generalized and applied to sustainable cities in general.

7.2 Research Findings and Contributions

7.2.1 Research Findings

An "Eco-city" can be regarded as the result of the globally reached consensus regarding paradigm shift towards sustainability in urban development. The focus, despite of its vagueness in terms and multifold interpretations, is to rebalance the economic development with ecological and environmental system and solve the "urban illnesses" in our society. Protecting the environmental conditions manifested as one of the core principles of this global trend for sustainable urbanization.

China's efforts of pursuing harmonious development between human and nature can be traced back to the oriental concept of "Shanshui City". From analyzing the Chinese Ecocity development, it is observed that China's Eco-cities aim to promote more sustainable growth regarding economic, environmental, and social aspects as national policies. The focus has begun to weigh in more towards the environment instead of economic development.

However, when compared to regional leading country of Japan, there are a number of categories out performed by the Japanese case in terms of efficiency and performance of energy sector, waste treatment and recycling sector, and pollution emission in particular. This indicates the challenges faced by such a geographically vast country like China with diverse local development levels and having one set of top-down national standards. More importantly, further improving and updating of the relevant guiding policies with more explicitly defined methods is definitely needed.

Low-carbon cities bring forth a more clear and measurable standard or principle "decarbonization" or "low-carbonization" for urban development. This global trend arose as a specific reaction towards the common foe of "climate change" and "global warming" faced by all human kind. And the development of the Kyoto Protocol in 1997 has globalized this "new norm" in urban sustainable developments.

China is currently the world's largest gross GHG emitting country, and Japan ranks the third in GHG emission after China and India in Asia (in 2013). To reduce GHG emissions, they have proposed and developed common but differentiate strategies. China focuses more on "CO2 emission reduction and adaptation plans" with strong government intervention and civil participation. While Japan's focus is on development of "low-carbon society" and "climate change resilience" that facilitated by sound science and technology applications.

Since late 1990s, many noticeable adjustments can be observed in China's urban development trajectory towards sustainability. The quick adaptation of low-carbonization as a national development objective reflects the international response to global warming and climate change. Additionally, a shift from top-down to bottom-up urban policy development approach is seen from the previous Garden City and Eco-city to Low-carbon City in China, due to its gigantic geographic coverage and local geopolitical characteristics.

However, there is still a lack of explicitness in definition or even overlapping in different urban development models. Sometimes, the low-carbon title or the political incentives behind these "environmentally-friendly city" designations outweigh the actual commitment in reducing carbon in the city developments. Also, there are many international practices that can offer excellent lessons or references to China's low-carbon city development.

The "smart cities" trend can be regarded as the most recent urban development model in the information era. With the rapid advancement of information and digitalization, ITCs such as big data and sensors, innovative means for gathering and processing information and data are being developed for city development and management, also known as "big-data based smart urban management". As the information and digitalization process accelerates in China, smart city development has been incorporated into the national development strategies that features in "the Internet plus extended industries" mode and ICT infrastructure development. While Japan's digitalization levels are higher than China, therefore Japanese smart city development has it own characters of goals, implementations and evaluation.

After reviewing and analyzing the current smart city concepts and frameworks, I conclude and propose a version conceptual framework for smart city as: "A city pursuing the twofold goal of improving quality of life while simultaneously realizing urban sustainability. The framework's major contents include both hard and soft urban domains such as "Governance", "People & Urban Living", "Economy", "Infrastructure", "Energy & Mobility", and "Environment". The active involvement of stakeholders and implementation of information and communication technologies (ICTs) are the major instruments for implementation."

It was then applied in a case study of Kitakyushu City of Japan using a customized Smart City Index consisting of 6 domains, 18 aspects with 36 measuring indicators. This was followed by an expert opinion survey to support an AHP method for indicator weighting. This additional step for indicator development proved to be very informative and useful in the selection of key performance indicators (KPIs) from numerous inventories and the prioritization of indicators for better or effective policy implementation.

In particular, the introduction of AHP as an indicator weighting mechanism or tool provides more rationality in the integration of indicator weights into the proposed index. Despite of certain limitations, the application of AHP in this thesis for expert survey provides some new insights of how "sustainable city" indicators could be further evaluated.

7.2.2 Research Contributions

The most important contribution of this thesis, besides the case studies and the existing body of literature regarding these urban development trends, is the conceptualized methodology (or methodological approach) from conceptual framework composition, to index or indicators selection and index development, to the final indicator weighting that has been distilled from these studies. It can be summarized as follows:

- First, defining the scopes and proposing a framework for urban/city development (i.e. eco-city, low-carbon city, smart city). This is achieved by having reliable sources of policy inputs such as those from the key decision makers or stakeholders, or from highly regarded literature.
- Second, developing an index or indicator system for measurement and evaluation based on the proposed framework. This required us firstly to identify indicator sources under specific goals or objectives, then to conduct a fine selection of supporting indicators based on SMART principles, and finally to integrate them into a complete index.
- Third, weighting or evaluating the proposed indicators using the Analytical Hierarchy Process (or other mathematical methods such as weighted preferences or Z score etc.). This can be realized by identifying expert groups for survey (could also be those people who are have expert knowledge of the intended areas) by questionnaires or workshops. And finally a weight is assigned to each individual indicator.

As an original contribution to the existing approaches for understanding and evaluating urban development, this integrated methodology will contribute to the pragmatic action that are flexible and adjustable with local context for policy and decision makers, urban managers, and key stakeholders from academia and industries.

7.3 Limitations

Urban sustainable development is a topic too broad to tackle or explain in a single dissertation. It can also be a simple term such as "development, which meets the needs of the present without compromising the ability of future generations to meet their own needs."

Therefore it is vital to consider and study it within proper local frames or boundaries. Due to the scope and research encompassed within this dissertation, there are several limitations regarding certain methods or process designs incorporated into the studies.

For the quantitative indicator comparisons of China's eco-cities with the case study from Japan, the baseline scenario was chosen from MHURD's standards. As is pointed out repeatedly throughout this thesis, various geo-political and geo-cultural situations can lead to different standards upheld and adopted in different regions or places. Not all the indicators are directly comparable, nor were the all the relevant data available. Even though Suzhou city is a very representative well-developed city in China, it can only represent Chinese cities with limited geo-social features.

For the low-carbon city research, a qualitative approach was taken to derive in-depth knowledge and references for improving China's low-carbon urban policies through field studies and in-depth interviews with key stakeholders in the selected cities of Japan and Germany. However, although the lessons learned from the two cases may offer some good references, they need to be presented to the key decision or policy makers in China before any concrete suggestions could be translated into reality. It would be helpful to have also selected two corresponding Chinese cities where the policies recommendations for low-carbonization could be compared, implemented and tested.

In regards to the smart city research, not like many studies where ICTs are given the most weight in consideration for either conceptualization or index development, I have considered it as a means to realize the urban development goals of improving life quality and pursing sustainability. But it is not my intention to claim that these ICTs like big data, sensors etc. are less important. On the contrary, the very core nature of smart cities are to utilize them for better urban goal or policy implementations. Thus the proposed smart city conceptual framework or index could lead to some disagreement due to the lack of ICT related contents in proportion. On the other hand, I also rigorously reiterate that locality needs to be integrated from the beginning in the policy design for smart cities.

The last search package of the integrated approach has improved upon the abovementioned limitations of the smart city study to certain extent. This integrated approach consists of a three-step mechanism with customizable sub-steps. But it is only based on one local case study of Kitakyushu City in Japan as of this time. For the expert option surveys, there are only 60 experts invited for this study, thus the sample size might not be that reliable. This also resulted from the adopted survey analytical tool of AHP, which has number limits in calculating the item evaluating inputs. For the questionnaire itself, some of feedback may be less accurate due to the unfamiliarity with the scale selections for some people. Improvement is definitely needed for such clarification in the instruction part of future surveys.

This integrated approach is summarized and distilled mainly from the three case studies enlisted in the thesis, therefore it might be confounding when applied to another model of sustainable urban or another different geographical setting. That is also the reason this approach needs to be flexible and customizable. And other major regret is that one of the main players of East Asia – South Korea was not included in this thesis, mainly due to the research scope and data availability. It may be very informative to see the smart city developing in Korea, given the high development in ICT infrastructures.

7.4 Future Research Perspectives

Ever since the industrial revolution, the momentum of human urbanization has continuously increased. A number of urban development trends have occurred along the timeline of human advancement during this time period. From Garden City and Green City, to Eco-City Low-carbon City, and to the current Smart City, all of them attempt to restore a balance between nature and the human sphere, between economic development and sustainability. The focus of each urban model differs, and there is not any proven "best" means or strategy that could meet the needs of all the parties involved. Under such a predicament, one effective tactic is to determine the most suitable way instead of the best way for preserving our common future in this urbanized era.

I have investigated several problems under the three major urban mega-trends in the recent decades, but there is more to be done. One topic that would be particularly interesting and worth further research, is to apply this proposed integrated approach to localization in different countries and regions. Only through application and feedback will this mechanism demonstrate its true value due to its high flexibility and customizability. Another topic that

could be further explored is the up scaling of the proposed SC index for measuring smart city performance in Japan. There are also several issues that could be further researched regarding the eco-/low-carbon cities in China, in terms of performance evaluation and policy recommendations with specific case study cities.

REFERENCES

- Akarte, M.M. (2001). "Web based casting supplier evalu- ation using analytic hierarchy process". Journal of the Operational Research Society, 52(5), 511-522.
- Albino, V., Berardi, U., & Dangelico, R.M. (2015). "Smart Cities: Definitions, Dimensions, Performance, and Initiatives". *Journal of Urban Technology*, 22(1), 3-21.
- Angelidou, M. (2014). "Smart city policies: A spatial approach". Cities, 41, S3-S11.
- Arcadis. (2015). Sustainable Cities Index 2015: Arcadis Corp.
- Awasthi, A., & Omrani, H. (2009). "A hybrid approach based on AHP and belief theory for evaluating sustainable transportation solutions". *International Journal of Global Environmental Issues*, 9(3), 212.
- Badri, M. (1999). "Combining the AHP and GP for global facility location-allocation problem". *International Journal of Production Economics*, 62(3), 237-248.
- Bakıcı, T., Almirall, E., & Wareham, J. (2012). "A Smart City Initiative: the Case of Barcelona". Journal of the Knowledge Economy, 4(2), 135-148.
- Bao, M. (2012). "Environmentalism and Environmental Movement in China since 1949. In McNeill,
 J. R. & Mauldin, E. S. (Eds.), A Companion to Global Environmental History. London: Blackwell Publishing Ltd.
- Barranco, R.R., Aurambout, J.-P., Silva, F.B.E., Herrera, M.M., Jacobs, C., & Lavalle, C. (2015). Indicators and trends for EU urban areas Joint Research Centre, Insistute for Environment and Sustainability.
- Barrionuevo, J.M., Berrone, P., & Ricart, J.E. (2012). "Smart Cities, Sustainable Progress". IESE Insight, 14, 50-57.
- Bayulken, B., & Huisingh, D. (2015). "A literature review of historical trends and emerging theoretical approaches for developing sustainable cities (part 1)". *Journal of Cleaner Production, 109*(2015), 11-24.
- Bennett, C.J. (2009). "What Is Policy Convergence and What Causes It?". *British Journal of Political Science*, *21*(02), 215.
- Bi, J., Zhang, R., Wang, H., Liu, M., & Wu, Y. (2011). "The benchmarks of carbon emissions and policy implications for China's cities: Case of Nanjing". *Energy Policy*, *39*(9), 4785-4794.
- Birkeland, J. (2012). "Design blindness in sustainable development: from closed to open systems design thinking". J. Urban Des., 17(2), 163-187.
- Boğaçhan, B. (2016). Toward a Theory of Successful Eco-Town Development: An integrative approach to characterizing and applying key 'success factors'. Erasums University Rotterdam, Nederlands. Retrieved from http://hdl.handle.net/1765/80098

- Bozbura, F.T., & Beskese, A. (2007). "Prioritization of organizational capital measurement indicators using fuzzy AHP". *International Journal of Approximate Reasoning*, 44(2), 124-147.
- Brenner, N., & Schmid, C. (2014). "The 'Urban Age' in Question". International Journal of Urban and Regional Research, 38(3), 731-755.
- Cagno, E., Caron, F., & Perego, A. (2001). "Multi-criteria assessment of the probability of winning in competitive bidding process". *International Journal of Production Management*, 19, 313-324.
- Calderoni, L., Maio, D., & Palmieri, P. (2012). "Location-aware Mobile Services for a Smart City: Desing, Implementation and Deployment". *Journal of theoretical and applied electronic commerce research*, 7(3), 15-16.
- Caragliu, A., Del Bo, C., & Nijkamp, P. (2011). "Smart Cities in Europe". Journal of Urban Technology, 18(2), 65-82.
- Cha, J., Ye, Z., & Tian, Y. (2014). "Current Status of International Low-carbon City Development. In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 38-62). Beijing: Scientific Publication
- Chen, J. (2007). "Rapid Urbanization in China: A real challange to soil protection and food security". *Catena*, 69(1), 1-15.
- Chen, T. (2010). "Smart grids, smart cities need better networks [Editor's Note". *IEEE Network*, 24(2), 2-3.
- Chen, W.Y. (2015). "The role of urban green infrastructure in offsetting carbon emissions in 35 major Chinese cities: A nationwide estimate". *Cities*, 44, 112-120.
- Chin, K.S., Chiu, S., & Tummala, V.M.R. (1999). "An evaluation of success factors using AHP to implement ISO 14001 based EMS". International Journal of Quality & Reliability Management, 16(4), 341-361.
- Chinese Society for Urban Studies. (2012). *China Low-Carbon Eco-City Development Report 2012*. Beijing: China Building Industry Press.
- Choi, J.H.-J. (2009). "The city is connections: Seoul as an urban network". *Multimedia Systems*, 16(1), 75-84.
- Chourabi, H., Nam, T., Walker, S., Gil-Garcia, J.R., Mellouli, S., Nahon, K., Pardo, T.A., & Scholl, H.J. (2012). "Understanding Smart Cities: An Integrative Framework". 2289-2297.
- Cocchia, A. (2014). "Smart and Digital City: A Systematic Literature Review. In Dameri, R. P. & Rosenthal-Sabroux, C. (Eds.), *Smart City* (pp. 13-43): Springer International Publishing.
- Cosgrave, E., Arbuthnot, K., & Tryfonas, T. (2013). "Living Labs, Innovation Districts and Information Marketplaces: A Systems Approach for Smart Cities". *Procedia Computer Science*, 16, 668-677.
- Cretu, L.-G. (2012). "Smart Cities Design Using Event-driven Paradigm and Semantic Web". *Informatica Economica*, 16(4), 57-67.

- De Montis, A., Toro, D., Droste-Franke, B., Omann, I., & Stagl, S. (2000, May 2000). *Criteria for quality assessment of MCDA methods*. Paper presented at the Proceedings of the 3rd Biennial Conference of the European Society for Ecological Economics, ESEE, Vienna, Austria.
- Deng, Z., & Ye, K. (2014). "Low-carbon Industry Develpment In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 201-215). Beijing: Scientific Publication
- Dewan, H. (2006). Sustainability Index. An Economic Perspective: Thompson Rivers University.
- Dhakal, S. (2002). Report on Indicators related research for Kitakyushu Initiative: Institute of Global Environmental Strategies.
- Dolowitz, D.P. (1997). "British Employment Policy in the 1980s: Learning from the American Experience". *Governance*, 10(1), 23-42.
- Economist Intelligence Unit. (2011). Asian Green City Index. Retrieved April 12, 2015, from Siemens <u>http://www.siemens.com/entry/cc/features/greencityindex_international/all/en/pdf/re</u> <u>port_asia.pdf</u>
- Edensor, T., & Jayne, M. (2012). "Introduction: urban theory beyond the West. In Edensor, T. & Jayne, E. (Eds.), Urban Theory Beyond the West: A World of Cites (pp. 1-27). London: Routledge.
- Eger, J.M. (2009). "Smart growth, smart cities, and the crisis at the pump a worldwide phenomenon". *I-Ways: The Journal of E-Government Policy and*
- Engwicht, D. (1993). Reclaiming Out Cities and Towns: Better Living With Less Traffic. Gabriola Island, BC: New Society Publishers.

European Commission. (1996). European Sustainable Cities Report. Brussels.

- Feng, Y.Y., & Zhang, L.X. (2012). "Scenario analysis of urban energy saving and carbon abatement policies: A case study of Beijing city, China". *Procedia Environmental Sciences*, 13, 632-644.
- Fleck, B. (2013). *Rhein-Hunsrück District From energy importer to energy exporter*. Paper presented at the 4th International Conference on Circular Economy, Birkenfeld, Germany.
- Fogliatto, F.S., & Albin, S.L. (2001). "A hierarchical method for evaluating products with quantitative and sensory char- acteristics". *IIE Transactions*, 33, 1081-1092.
- Fook, L.L., & Gang, C. (2010). "Towards Eco-Cities in East Asia. In Fook, L. L. & Gang, C. (Eds.), *Towards A Livable And Sustainable Urban Environment: Eco-Cities in East Asia* (pp. 219). Singapore: World Scientifc Publishing.
- Forgionne, G.A., & Kohli, R. (2001). "A multi-criteria assessment of decision technology system and journal quality". *Information and Management*, 38(7), 171-183.
- Giffinger, R., Fertner, C., Kramar, H., Kalasek, R., Pichler-Milanovic, N., & Meijers, E. (2007a). Smart cities-Ranking of European medium-sized cities: Vienna University of Technology.

- Giffinger, R., Fertner, C., Kramar, H., Kalasek, R., Pichler-Milanovic, N.A., & Meijers, E. (2007b). Smart cities-Ranking of European medium-sized cities: Vienna University of Technology.
- Giffinger, R., & Gudrun, H. (2010). "Smart cities ranking: an effective instrument for the positioning of the cities?". ACE: Architecture, City and Environment, 4(12), 7-26.
- Girardet, H. (2003). "Making Adelaide a Green City Adelaide Thinkers in Residence Inaugural Public Lecture (pp. 1e13.).
- Glaeser, E.L. (2011). Tirumph of the City. London: Macmillan.
- Godfrey, L., & Todd, C. (2001). Defining thresholds for freshwater sustain- ability indicators within the context of South African water resource management. Paper presented at the 2nd WARFA/Waternet Symposium: Integrated Water Resource Management: Theory, Practice, Cases., Cape Town, South Africa. http://www.waternetonline.ihe.nl/aboutWN/pdf/godfrey.pdf.
- Goepel, K.D. (2013). BPMSG AHP Excel Template with multiple Inputs. Retrieved June 28, 2015
- Graedel, T.E. (1999). "Industrial Ecology and the Ecocity". The Bridge, 29(4), 10-14.
- Guan, L. (2012). "Smart Steps To A Battery City". Government News, 32(2), 24-27.
- Hall, P. (1998). Cities in Civilization. New York: Pantheon.
- Hall, P.A. (1993). "Policy Paradigms, Social Learning, and the State: The Case of Economic Policymaking in Britain". *Comparative Politics*, 25(3), 275.
- Hall, R.E. (2000). The Vision of A Smart City.
- Harbi, K.M.A. (2001). "Application of AHP in project management". International Journal of Project Management, 19(4), 19-27.
- Harrison, C., Eckman, B., Hamilton, R., Hartswick, P., Kalagnanam, J., Paraszczak, J., & Williams,
 P. (2010). "Foundations for Smarter Cities". *IBM Journal of Research and Development*, 54(4), 1-16.
- Holden, M., Roseland, M., Ferguson, K., & Perl, A. (2008). "Seeking urban sustainability on the world stage". *Habitat International*, 32(3), 305-317.
- Hollands, R.G. (2008). "Will the real smart city please stand up?". City, 12(3), 303-320.
- Howard, E. (1898). *To-morrow: A Peaceful Path to Real Reform*. London: Swan Sonnenschein & Co.
- Huang, G. (1989). *Garden Cities, Green Cities, Eco-Cities*. Chongqing: nstitute of Urban Planning, Construction University of Construction Engineering.
- Huang, Q., & Yang, D. (2001). "Study Reviews on Eco-city Theories Domestic and Abroad". Urban Planning, 25(1), 59-66.
- Ida. (2012). iN2015 Masterplan. Singapore.

- Idowu, S., & Bari, N. (2012). A Development Framework for Smart City Services. (MSc.), Lulea University of Technology, Lulea, Sweden.
- Imura, H. (2010). "Eco-Cities: Re-Examining Concepts and Approaches. In Liang, F. L. & Gang, C. (Eds.), Towards a Livable and Sustainable Urban Environment: Eco-cities in East Asia: World Scientific Publishing Company.
- Institute for Global Environmental Strategies. (2005). Asia-Pacific Environmental Innovation Strategies (APEIS)
- Research on Innovative and Strategic Policy Options (RISPO)
- Good Practices Inventory: KItakyushu Eco-Town Project. online: Retrieved from http://pub.iges.or.jp/contents/APEIS/RISPO/inventory/db/pdf/0147.pdf.
- International Orgnization for Stadardization. (2014). ISO/IEC JTC 1 Smart Cities. Switzerland: ISO/IEC 2015.
- Ishida, T., & Isbister, K. (2000). Digital Cities: Experiences, Technologies and Future Perspectives. Berlin: Springer-Verlag.
- Jiang, P., & Tovey, N.K. (2009). "Opportunities for low carbon sustainability in large commercial buildings in China". *Energy Policy*, 37(11), 4949–4958.
- Jong, M.D., Joss, S., Schraven, D., Zhan, C., & Weijnen, M. (2015). "Sustainableesmarteresilientelow carboneecoeknowledge cities; making sense of a multitude of concepts promoting sustainable urbanization". *Journal of Cleaner Production*, 109(2015), 25-38.
- Joss, S. (2011). "Eco-Cities: The Mainstreaming of Urban Sustainability: Key Characteristics and Driving Factors". *International Journal of Sustainable Development and Planning*, 6(3), 268-269.
- Joss, S. (2012). Tomorrow's City Today, Eco-city Indicators, Standards and Frameworks. In Joss, S. (Ed.), *Bellagio Conference Report*. London: University of Westminster.
- Joss, S., & Tomozeiu, D. (2013). 'Eco-City' Frameworks A Global Review. Online: University of Westminster.
- Joss, S., Tomozeiu, D., & Cowley, R. (2011). Eco Cities A Global Survey 2011, University of Westminster International Eco-Cities Initiatives, London. London: International Eco-Cities Initiatives.
- Kahri, F., Williams, J., Ding, J., & Hu, J. (2011). "Challenges to China's transition to a low carbon electricity system". *Energy Policy*, *39*(7), 4032-4041.
- Kei. (2005). Knowledge Economy Indicators, Work Package 7, State of the Art Report on Simulation and Indicators.
- Kengpol, A., & O'brien, C. (2001). "The development of a decision support tool for the selection of advanced technology to achieve rapid product development". *International Journal of Production Economics*, 69(2), 177-191.

- Khakee, A. (2002). "Assessing institutional capital building in a Local Agenda 21 process in Goeteborg.". *Plan. Theory Pract.*, 3(1), 53-68.
- Khanna, N., Fridley, D., & Hong, L. (2014). "China's pilot low-carbon city initiative: A comparative assessment of national goals and local plans". *Sustainable Cities and Society*, 12(12), 110-121.
- Kil, S.-H., Lee, D., Kim, J.-H., Li, M.-H., & Newman, G. (2016). "Utilizing the Analytic Hierarchy Process to Establish Weighted Values for Evaluating the Stability of Slope Revegetation based on Hydroseeding Applications in South Korea". Sustainability, 8(1), 58.
- Komninos, N. (2009). "Intelligent cities: towards interactive and global innovation environments". International Journal of Innovation and Regional Development, 1(4), 337-355.
- Komninos, N. (2011). "Intelligent cities: Variable geometries of spatial intelligence". *Intelligent Buildings International*, 3(3), 172-188.
- Kourtit, K., Macharis, C., & Nijkamp, P. (2014). "A multi-actor multi-criteria analysis of the performance of global cities". *Applied Geography*, 49, 24-36.
- Kourtit, K., Nijkamp, P., & Arribas, D. (2012). "Smart cities in perspective a comparative European study by means of self-organizing maps". *Innovation: The European Journal of Social Science Research*, 25(2), 229-246.
- Kwon, O., & Kim, J. (2007). "A methodology of identifying ubiquitous smart services for U-City development. In J., I., Yang, L. T., Cao, J., Ma, J. & T. Ungerer (Eds.), Ubiquitous Intelligence and Computing Proceedings (pp. 143-152). Berlin: Springer.
- Lai, V.S., Wong, B.K., & Cheung, W. (2002). "Group decision making in a multiple criteria environment: A case using the AHP in software selection". *European Journal of Operational Research*, 137(1), 134-144.
- Langhelle, O. (2000). "Why ecological modernization and sustainable development should not be conflated". J. Environ. Policy & Plan., 2(4), 303-233.
- Larmour, P. (2002). "Policy transfer and reversal: customary land registration from Africa to Melanesia". *Public Administration and Development*, 22(2), 151-161.
- Lazaroiu, G.C., & Roscia, M. (2012). "Definition methodology for the smart cities model". *Energy*, 47(1), 326-332.
- Lee, J.H., Hancock, M.G., & Hu, M.-C. (2014). "Towards an effective framework for building smart cities: Lessons from Seoul and San Francisco". *Technological Forecasting and Social Change*, 89, 80-99.
- Lehmann, S. (2013). "Low-to-no carbon city: Lessons from western urban projects for the rapid transformation of Shanghai". *Habitat International*, *37*(37), 61-69.
- Li, Y., Lin, Y., & Geertman, S. (2015). *The development of smart cities in China*. Paper presented at the The 14th International Conference on Computers in Urban Planning and Urban Management, Cambridge, MA USA.

- Li, Y., & Qiu, L. (2015). "A comparative study on the quality of China's eco-city: Suzhou vs Kitakyushu". *Habitat International*, 50(2015), 57-64.
- Li, Z., Chang, S., Ma, L., Liu, P., Zhao, L., & Yao, Q. (2012). "The development of low-carbon towns in China: Concepts and practices". *Energy Policy*, 47(1), 590-599.
- Li, Z., Chang, S., Ma, L., Liu, P., Zhao, L., & Yao, Q. (2012). "The development of low-carbon towns in China: Concepts and practices". *Energy*, 47(1), 590-599.
- Litfin, K. (2014). Eco-villages: Lessons for sustainable community. Cambridge, UK: Polity Press.
- Liu, W., & Qin, B. (2016). "Low-carbon city initiatives in China: A review from the policy paradigm perspective". *Cities*, *51*, 131-138.
- Lombardi, P. (2011). "New challenges in the evaluation of Smart Cities
- ". NETWORK INDUSTRIES QUARTERLY, 13(3), 8-10.
- Lombardi, P., Giordano, S., Farouh, H., & Yousef, W. (2012). "Modelling the smart city performance". Innovation: The European Journal of Social Science Research, 25(2), 137-149.
- Lv, F. (2012). "Smart City Development Level Assessment for Tianjin Using AHP and Gray Comprehensive Evaluation". Advances in information Sciences and Service Sciences, 4(19.10), 74-82.
- Maeda, T. (2010). Kitakyushu Initiative for a Clean Environment Final Report. In Secretariat of the Kitakyushu Initiative Network (Ed.). Kanagawa, Japan: Institute for Global Environmental Strategies (IGES).
- Mahizhnan, A. (1999). "Smart cities The Singapore case". Cities, 16(1), 13-18.
- Marsal-Llacuna, M.-L., Colomer-Llinàs, J., & Meléndez-Frigola, J. (2015). "Lessons in urban monitoring taken from sustainable and livable cities to better address the Smart Cities initiative". *Technological Forecasting and Social Change*, *90*, 611-622.
- Meadows, D.H., Meadows, D.L., Rangers, J., & Behrens, W.W. (1972). Limits to growth. A report to the club of Rome. New York.
- Mebratu, D. (1998). "Sustainability and Sustainable Development Historical and Conceptual Review". *Environ. Impact Asses. Rev., 18,* 493-530.
- Mega, V., & Pedersen, J. (1998). Urban Sustainability Indicators. Dublin, Ireland: European Foundation for the Improvement of Living and Working Conditions.
- Mickwitz, P., Hildéna, M., Seppälä, J., & Melanena, M. (2011). "Sustainability through system transformation: lessons
- from Finnish efforts". Journal of Cleaner Production, 19(16), 1779-1787.
- Ministry of Environment Protection. (2004). Eco-County, Eco-Province Index. from MEP http://www.zhb.gov.cn/gkml/zj/wj/200910/t20091022_172492.htm

- Ministry of Finance of the People's Republic of China. (2011). Notice on Implementation of Lowcarbon Demonstration Town Pilot Program. online: Retrieved from http://jjs.mof.gov.cn/zhengwuxinxi/tongzhigonggao/201106/t20110616_562707.html
- Ministry of Housing Urban and Rural Development. (2004). Eco-Garden City Standards and Selection Criteria. from MoHURD http://www.mohurd.gov.cn/zcfg/jsbwj 0/jsbwjcsjs/201212/t20121207 212220.htm
- Mitton, N., Papavassiliou, S., Puliafito, A., & Trivedi, K.S. (2012). "Combining Cloud and sensors in a smart city environment". *EURASIP JOURNAL ON WIRELESS COMMUNICATIONS AND NETWORKING*, 2012(1), 247.
- Myers, G. (2014). "From expected to unexpected comparisons: Changing the flows of ideas about cities in a postcolonial urban world". *Singapore Journal of Tropical Geography*, 35(1), 104-118.
- Nam, T., & Pardo, T.A. (2011a). Conceptualizing Smart City with Dimensions of Technology, People, and Institutions. Paper presented at the 12th Annual International Conference on Digital Government Research, MD, USA.
- Nam, T., & Pardo, T.A. (2011). Conceptualizing smart city with dimensions of technology, people, and institutions. Paper presented at the Proceedings of the 12th Annual International Digital Government Research Conference: Digital Government Innovation in Challenging Times.
- Nam, T., & Pardo, T.A. (2011b, Septermber 26-28, 2011). Smart City as Urban Innovation: Focusing on Management, Policy, and Context. Paper presented at the 5th International Conference on Theory and Practice of Electronic Governance, Tallinn, Estonia.
- National Development and Reform Commission. (2011). National Development and Reform Commission's Notice on the Second Batch of Trial Low-carbon Province and Low-carbon City. Retrieved from http://qhs.ndrc.gov.cn/gzdt/201212/t20121205 517419.html.
- National People's Congress. (2011). The Twelfth Five-Year Plan on national
- economy and social development. Online: Retrieved from http://www.gov.cn/2011lh/content 1825838 2.htm.
- Ness, B., Urbel Piirsalu, E., Anderberg, S., & Olsson, L. (2007). "Categorising tools FOS sustainability assessment". *Ecological Economics*, 60(498-508).
- Oecd. (2013). Green Growth in Kitakyushu, Japan OECD Green Growth Studies.
- Ong, A., & Roy, A. (2011). In Ong, A. & Roy, A. (Eds.), Worlding Cities: Asian Experiments in the Art of Being Global. Oxford: Wiley-Blackwell.
- Paskaleva, K.A. (2011). "The smart city: A nexus for open innovation?". Intelligent Buildings International, 3(3), 153-171.
- Patel, S. (2014). "Is there a 'south' perspective to urban studies? In Parnell, S. & Oldfield, S. (Eds.), *The Routledge Handbook on the Cities of the Global South* (pp. 37-53). London: Routledge.

- Perboli, G., De Marco, A., Perfetti, F., & Marone, M. (2014). "A New Taxonomy of Smart City Projects". *Transportation Research Procedia*, 3, 470-478.
- Perera, C., Zaslavsky, A., Christen, P., & Georgakopoulos, D. (2014). "Sensing as a service model for smart cities supported by Internet of Things". *TRANSACTIONS ON EMERGING TELECOMMUNICATIONS TECHNOLOGIES*, 25(1), 81-93.
- Pierson, C. (1997). "Learning from labour? Welfare policy transfer between Australia and Britain". *Commonwealth & Comparative Politics, 41*(1), 77-100.
- Qi, Y., & Wu, T. (2013). "The politics of climate change in China". *Wiley Interdisciplinary Reviews: Climate Change*, 4(4), 301-313.
- Qin, B., & Han, S.S. (2013). "Planning parameters and household carbon emission: Evidence from high and low-carbon neighborhoods in Beijing". *Habitat International*, *37*(1), 52-60.
- Qiu, B. (2010). Ideals and Realities Initial Investigation on Chinese Low-Carbon Cities Indicator System Development and Practice. Beijing: China Industrial Development Press.
- Quental, N., LourençO, J.M., & Da Silva, F.N. (2009). "Sustainable development policy: goals, targets and political cycles". *Sustain. Dev.*, 19(1), 15-29.
- Ramachandran, N. (2000). *Monitoring Sustainability: Indices and Techniques of Analysis*. New Delhi: Concept Publishing Company.
- Ramanathan, R., & Ganesh, L.S. (1995). "Using AHP for resource allocation problems". *European Journal of Operational Research*, 80(2), 410-417.
- Register, R. (1987). *Ecocity Berkeley: Building Cities For a Healthy Future*. Berkeley: North Atlantic Books.
- Register, R. (2006). *Eco-cities: Rebuilding Cities in Balance with Nature*. Canada: New Society Publishers.
- Roseland, M. (1997). "Dimensions of Eco-cities". Cities, 14(4), 197-198.
- Saaty, T.L. (1980). The analytic hierarchy process. New York: McGraw-Hill.
- Schuurman, D., Baccarne, B., & De Marez, L. (2012). "Smart Ideas for Smart Cities: Investigating Crowdsourcing for Generating and Selecting Ideas for ICT Innovation in a City Context". *Journal of theoretical and applied electronic commerce research*, 7(3), 11-12.
- Scott, A.J., & Storper, M. (2015). "The Nature of Cities: The Scope and Limits of Urban Theory". International Journal of Urban and Regional Research, 39(1), 1-15.
- Shapiro, J.M. (2006). "Smart Cities: Quality of Life, Productivity, and the Growth Effects of Human Capital". *Review of Economics and Statistics*, 88(2), 324-335.
- Shelton, T., Zook, M., & Wiig, A. (2015). "The 'actually existing smart city'". Cambridge Journal of Regions, Economy and Society, 8(1), 13-25.
- Shen, Q. (1998). Urban Ecology and Urban Environment. Shanghai: Tongji University Publication.

- Shen, Q., An, C., & Yan, G. (2014). "International Comparison of Low-carbon City Standards. In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 62-140). Beijing: Scientific Publication
- Sheppard, E., Leitner, H., & Maringanti, A. (2013). "Provincializing Global Urbanism: A Manifesto". Urban Geography, 34(7), 893-900.
- Singh, R.K., Murty, H.R., Gupta, S.K., & Dikshit, A.K. (2012). "An overview of sustainability assessment methodologies". *Ecological Indicators*, 15(1), 281-299.
- Song, Y. (2011). "Ecological City and Urban Sustainable Development". *Procedia Engineering*, 21, 142-146.
- State Council of the People's Republic of China. (2009). China to cut 40 to 45% GDP unit carbon by2020.Online:Retrieved fromhttp://www.gov.cn/zwgk/2010-05/05/content_1599897.htm.
- Storper, M. (2016). "Current Debates in Urban Theory: A Critical Assessment (Forthcoming)". Urban Studies.
- Su, M., Xu, L., Chen, B., & Yang, Z. (2013). "Eco-City Planning Theories and Thoughts. In Yang, Z. (Ed.), *Eco-Cities: A Planning Guide*. (pp. 5-6). FL: CRC Press.
- Su, M.R., Chen, B., Xing, T., Chen, C., & Yang, Z.F. (2012). "Development of low-carbon city in China: Where will it go?". *Procedia Environmental Sciences*, 13, 1143-1148.
- Sun, Y. (2014). Low-Carbon Model Town Project Development in China. Paper presented at the APEC Forum.
- Suzhou City. (2015). Introduction of Suzhou (in Chinese). Online.
- Thite, M. (2011). "Smart cities: implications of urban planning for human resource development". *Human Resource Development International*, 14(5), 623-631.
- Thuzar, M. (2011). "URBANIZATION IN SOUTHEAST ASIA: DEVELOPING SMART CITIES FOR THE FUTURE? In Montesano, M. J. (Ed.), *Regional Outlook: Southeast Asia 2011-2012* (pp. 96-100): Institute of Southeast Asian Studies.
- Tong, O., Shao, S., Zhang, Y., Chen, Y., Liu, S.L., & Zhang, S.S. (2012). "An AHP-based waterconservation and waste-reduction indicator system for cleaner production of textile-printing industry in China and technique integration". *Clean Technologies and Environmental Policy*, 14(5), 857-868.
- Tummala, V.M.R., Chin, K.S., & Ho, S.H. (1997). "Assessing success factors for implementing CE a case study in Hong Kong electronics industry by AHP". *International Journal of Production Economics*, 49(3), 265-283.
- Un-Habitat. (2011). Global Report on Human Settlements 2011: Cities and Climate Change.: Earthscan Ltd.
- United Nations. (2007). Indicators of Sustinable Development: Guidelines and Methodologies (Thrid Edition ed.). New York.

- United Nations. (2012). World Population Prospects. Retrieved June 06, 2014, from United Nations http://esa.un.org/unpd/wpp/unpp/p2k0data.asp
- United Nations. (2015). World Urbanization Prospects: The 2014 Revision, . UN Website: Department of Economic and Social Affairs, Population Division.
- Vaidya, O.S., & Kumar, S. (2004). "Analytic hierarchy process: An overview of applications". *Euopean Jounal of Operational Research, 169*(2006), 1-29.
- Vanolo, A. (2013). "Smartmentality: The Smart City as Disciplinary Strategy". Urban Studies, 51(5), 883-898.
- Wang, M., Zhang, L., & Gao, K. (2014a). "Current Status on China's Low-carbon City Development In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 13-36). Beijing Scientific Publication
- Wang, M., Zhang, L., & Gao, K. (2014b). "Indicator Systems of China's Low-carbon City In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 251-312). Beijing: Scientific Publication.
- Wang, R. (2000). "Progress of the Urban Ecology Studies in the Transformation Age". Journal of Ecology, 20(5), 830-840.
- Wang, X., Li, X., Zhen, F., & Zhang, J. (2016). "How smart is your tourist attraction?: Measuring tourist preferences of smart tourism attractions via a FCEM-AHP and IPA approach". *Tourism Management*, 54, 309-320.
- Wang, X.R. (2001). "On the theories, ways and counter measures for the construction of eco-city A case study of Shanghai, China". *Journal of Fudan University (Natural Science)*, 40(4), 349-354.
- Washburn, D., Sindhu, U., Balaouras, S., & Dines, R.A. (2009). "Helping CIOs Understand "Smart City" Initiatives". *Growth*.
- World Bank. (2009). Sino-Singapore Tianjin Eco-city: A Case Study of an Emerging Eco-City in China Technical Assistance (TA) Report.
- World Commission on Environment and Development. (1987). Our common future. New York.
- Wu, X., & Yang, Z. (2010). "The Concept of Smart City and Future City Development [J]". Urban Studies, 11(011).
- Xi, F., Y., G., X., C., Y., Z., X., W., B., X., & Zhu, Q. (2011). "Contributing to local policy making on GHG emission reduction through inventorying and attribution: A case study of Shenyang, China". *Energy Policy*, 39(10), 5999–6010., 39(10), 5999-6010.
- Xie, P.F., Zhou, L.L., Liu, Y., Zhang, A.H., Pang, T., & Song, X.F. (2010). "Eco-city Indicator System Construction and Eco-city Demonstration Evaluation". *Urban Studies*, 17(7), 12.
- Yang, L., & Li, Y. (2013). "Low-carbon City in China". Sustainable Cities and Society, 9, 62-66.
- Yanitsky, O. (1981a). "Cities and Human Ecology Social Problems of Man's Environment: Where We Live and Work. Moscow: Progress Publishers.

- Yanitsky, O. (1981b). "Social Problems of Man's Environment: Where We Live and Work. Moscow: Progress Publishers *Cities and Human Ecology*. Moscow: Progress Publishers.
- Yu, L. (2009). "Study on development objectives and implementing policies of Chinese eco-city". *Urban Planning International, 6*, 102-107.
- Yu, L. (2014). "Low carbon eco-city: New approach for Chinese urbanisation". *Habitat International, 44*(2014), 102-110.
- Zelda, B. (2009). "Industry and the smart city". Dissent, 56(3), 27-34.
- Zhang, L., Feng, Y., & Chen, B. (2012). "Alternative scenarios for the development of a low- carbon city: A case study of Beijing". *China. Energy*, 4(2).
- Zhang, S., Kuang, X., Chen, Y., Deng, X., & Chen, J. (2014). "Low-carbon City Planning and Design. In Huang, W. & Wang, J. (Eds.), *China Low-carbon City Construction Report* (pp. 143-199). Beijing: Scientific Publication
- Zhang, Y., & Du, Z. (2011). "Smart City Construction Status in China". China Information Times, 168(2), 28-32.
- Zhang, Z. (2010). "China in the transition to a low-carbon economy". *Energy Policy*, 38(11), 6638-6653.
- Zhou, N., He, G., & Williams, C. (2012). China's Development of Low-Carbon Eco-Cities and Associated Indicator System.
- Zhou, N., & Williams, C. (2013). An International Review of Eco-City Theory, Indicators, and Case Studies: China Energy Group, Environmental Energy Technologies Division, Lawrence Berkeley National Laboratory.
- Zhou, Y., Zhang, B., Zou, J., Bi, J., & Wang, K. (2012). "Joint R&D in low-carbon technology development in China: A case study of the wind-turbine manufacturing industry". *Energy Policy*, 46(100-108).
- Zou, X. (2015a). "Book Review: Chinese Low-Carbon City Construction Report". Asia Pacific World, 6(1), 93-95.
- Zou, X. (2015b). "Book Review: Ecovillages: Lessons for Sustainable Community. By Karen T. Litfin". Asia Pacific World, 6(1), 100-101.
- Zou, X., & Li, Y. (2014). "How Eco are China's Eco-Cities? An International Perspective". International Review for Spatial Planning and Sustainable Development, 2(3), 18-30.
- Zou, X., & Li, Y. (2015). "Developing Tailor-Made Urban Environmental Policies for China's Low-Carbon Cities—Implications from Japan and Germany. In Feng, S., Huang, W., Wang, J., Wang, M. & Zha, J. (Eds.), Low-carbon City and New-type Urbanization Proceedings of Chinese Low-carbon City Development International Conference (pp. 273-284): Springer Berlin Heidelberg.
- Zou, Z.H., Yun, Y., & Sun, J.N. (2006). "Entropy method for determination of weight of evaluating indicators in fuzzy synthetic evaluation for water quality assessment". J Environ Sci (China), 18(5), 1020-1023.

Zygiaris, S. (2012). "Smart City Reference Model: Assisting Planners to Conceptualize the Building of Smart City Innovation Ecosystems". *Journal of the Knowledge Economy*, 4(2), 217-231.

APPENDIX

Table A 1 Smart City Index Weighting Inputs Breakdown

SC Governance			P 1	P 2	Р 3	P 4	P 5	P 6	Р7	P 8	AW	Rankings
Perception of tra	nsparency of bureaucracy		7.2%	7.0%	49.0%	44.0%	17.0%	52.0%	45.0%	37.0%	32.3%	1
Perception of fig	ht against corruption		13.6%	12.0%	10.0%	21.0%	6.0%	4.0%	22.0%	15.0%	13.0%	4
Monitoring environmental performance			38.2%	39.0%	18.0%	22.0%	48.0%	17.0%	3.0%	8.0%	24.2%	2
City representatives per (1000) residents			3.5%	4.0%	4.0%	7.0%	11.0%	16.0%	3.0%	3.0%	6.4%	5
Female city representatives per (1000) residents			6.1%	7.0%	4.0%	3.0%	8.0%	4.0%	4.0%	8.0%	5.5%	6
Public participat	ion in environmental decision-	making	31.4%	32.0%	15.0%	3.0%	11.0%	8.0%	24.0%	29.0%	19.2%	3
GCR	67.70%	CR	11.7%	9.0%	9.0%	6.0%	9.0%	8.0%	10.0%	9.0%		
SC Economy		P 1	P 2	P 3	P 4	P 5	P 6	Р7	P 8	AW	Rankings	
% of budget of local government allocated for environment			5.0%	24.0%	27.0%	5.0%	28.0%	45.0%	12.0%	11.0%	19.6%	3
R&D expenditur	re in % of GDP		12.0%	42.0%	47.0%	21.0%	2.0%	19.0%	19.0%	12.0%	21.8%	1

Use of electricity	y per GDP		16.0%	8.0%	7.0%	13.0%	42.0%	16.0%	23.0%	38.0%	20.4%	2
Use of water per GDP			16.0%	10.0%	6.0%	28.0%	4.0%	12.0%	10.0%	20.0%	13.3%	5
Gross city produ	et per capita		27.0%	10.0%	9.0%	21.0%	17.0%	5.0%	13.0%	6.0%	13.5%	4
Households belo	w poverty line		23.0%	6.0%	4.0%	11.0%	7.0%	3.0%	23.0%	14.0%	11.4%	6
GCR	57.20%	CR	4.0%	8.0%	7.0%	8.0%	10.0%	10.0%	10.0%	9.0%		
SC People &Urban Living			P 1	P 2	P 3	P 4	P 5	P 6	Р 7	P 8	AW	Rankings
Number of doctors per 1000 population			38.0%	36.0%	6.0%	6.0%	50.0%	19.0%			25.8%	1
Number of hosp	itals per 1000 population		5.0%	17.0%	4.0%	6.0%	18.0%	3.0%			8.8%	6
Number of envir	conmental staffs in city govern	ment per 1000										
population			11.0%	9.0%	19.0%	5.0%	2.0%	10.0%			9.3%	5
% of industries compliant with emission control regulations			9.0%	4.0%	32.0%	38.0%	9.0%	10.0%			17.0%	3
% of vehicles compliant with emission control regulations			11.0%	7.0%	29.0%	17.0%	6.0%	10.0%			13.3%	4
Adult literacy ra	te		26.0%	28.0%	10.0%	27.0%	15.0%	47.0%			25.5%	2

GCR	74.00%	CR	6.0%	10.0%	8.0%	10.0%	10.0%	10.0%				
SC Infrastructure			P 1	P 2	Р 3	P 4	Р 5	P 6	Р 7	P 8	AW	Rankings
Energy consump	otion of residential buildings		22.0%	17.0%	11.0%	7.0%	11.0%	56.0%			20.7%	3
Energy-efficient building standards				16.0%	29.0%	17.0%	9.0%	3.0%			14.7%	4
Green spaces per capita				4.0%	6.0%	64.0%	70.0%	27.0%			38.3%	1
Accessibility of	smart grid		5.0%	64.0%	53.0%	12.0%	10.0%	14.0%			26.3%	2
GCR	68.00%	CR	9.0%	10.0%	9.0%	7.0%	10.0%	10.0%				
	SC Energy & Mobility		P 1	P 2	Р 3	P 4	P 5	P 6	Р 7	P 8	AW	Rankings
Share of renewal	ble energy in total energy use		41.0%	69.0%	28.0%	30.0%	50.0%	59.0%	41.0%	56.0%	46.8%	1
CO2 per capita f	rom energy use		46.0%	10.0%	48.0%	12.0%	6.0%	25.0%	21.0%	31.0%	24.9%	2
Green mobility s	share		6.0%	7.0%	12.0%	35.0%	22.0%	12.0%	15.0%	4.0%	14.1%	3
E-vehicle in com	nmercial vehicle shares		6.0%	15.0%	12.0%	23.0%	22.0%	4.0%	23.0%	8.0%	14.1%	3

GCR	79.10%	CR	1.0%	9.0%	7.0%	8.0%	6.0%	9.0%	8.0%	10.0%		
SC Environment			P 1	P 2	Р 3	P 4	P 5	P 6	Р 7	P 8	AW	Rankings
Air quality (indicated by SO2, Total Suspended Particles etc.)				4.0%	31.0%	44.0%	3.0%	13.0%	19.0%	30.0%	19.4%	2
% of population	with access to adequate and c	lean water	5.0%	3.0%	4.0%	21.0%	37.0%	51.0%	48.0%	35.0%	25.5%	1
Water quality (measured by BOD, COD contents etc.)				8.0%	23.0%	22.0%	16.0%	7.0%	19.0%	20.0%	15.8%	4
Per capita waste	generation		42.0%	15.0%	9.0%	7.0%	4.0%	4.0%	3.0%	5.0%	11.1%	5
% of total munic	ipal solid waste (MSW) colled	cted & treated	21.0%	22.0%	9.0%	3.0%	6.0%	13.0%	5.0%	5.0%	10.5%	6
% of total wastewater treated or recycled			10.0%	48.0%	24.0%	3.0%	34.0%	13.0%	5.0%	5.0%	17.8%	3
GCR	57.60%	CR	9.0%	10.0%	8.0%	6.0%	10.0%	6.0%	9.0%	8.0%		

Note: "P" stands for "Participant"

"GCR" stands for "Group Consensus Rate"

"CR" stands for "Consistent Ratio"

"AW" stands for "Aggregated Weight"

Table A 2 Expert Survey Questionnaires



				Your	Inputs	Below					
	Wh Sm	ich indica	ator do yo in the as	u think is pect of " <mark>G</mark>	more Ir overna	nportant nce"? (Cł	for meas noose on	suring a le at each	City's 1 row)		
		Indic	ators (A)		vs		Indica	ators (B)			
	Transpar	ency of bu	eaucracy		Vs	Fight agai	inst corrup	tion			
1	e 🗌		□5	□3			5	7	9		
_	Transpar	ency of bui	reaucracy		γ_s	Monitor i	ts environ	mental per	formances		
2	9		□5	□3	\square_1				9		
_	Transparency of bureaucracy				γ_s	City repre	esentatives	s per (1000)	resident		
3	9		□5	□3	\square_1				9		
	Transpar	ency of bu	eaucracy		Vs	Female ci resident	ty represe	ntatives pe	r (1000)		
4	9	7	□5	□3			□5	7	9		
	Transpar	ency of bu	eaucracy		γs	Public par decision-	rticipation making	in environr	nental		
5	9		□5				5		9		
	Fight aga	inst corrup	tion		γs	Monitor i	ts environ	mental perf	formances		
6	9	7	□5	□3	\Box_1		5	7	9		
	Fight aga	inst corrup	tion		γs	City representatives per (1000) resident					
7	9		□5	□3		3	5	7	9		
	Fight aga	inst corrup	tion	1	νs	Female ci resident	ty represe	ntatives pe	r (1000)		
8	9	7	□5	□3	\Box_1		□5	7	9		
	Fight aga	inst corrup	tion		Vs	Public par decision-	rticipation making	in environr	nental		
9	9	7	□5	□3	\Box_1		5	7	9		
	Monitor	its environ	mental perf	ormances	Vs	City repre	esentatives	per (1000)	resident		
10	9	7	□5	□3	\Box_1		5	7	9		
	Monitor	its environ	mental perf	ormances	Vs	Female ci resident	ty represe	ntatives pe	r (1000)		
11	9	7	□5	□3	\Box_1		□5		9		
	Monitor	its environ	mental perf	ormances	Vs	Public par	rticipation	in environr	nental		
12	9	7	□5	□3			5	7	9		
	City repr	esentatives	per (1000)	resident	Vs	Female ci	ty represe	ntatives pe	r (1000)		
13	9	7	□5	□3			5	7	9		
	City repr	esentatives	per (1000)	resident	Vs	Public par decision-	rticipation making	in environr	nental		
14	9	7	5	3				D 7	9		
	Female c	ity represe	ntatives pe	r (1000)	Vs	Public par	rticipation	in environr	nental		
15		D 7	 5	□3				— 7	e		
		2	Please Le	ave your	commer	ıts, íf thei	re ís any				
						~					

				Your I	nputs	Below	,			Return to
	Whi Si	ch indica martnes	ator do you s in the as	u think is spect of "I	more Ir Econor	nportant ny"? (Ch	for meas oose one	at each r	<mark>City's</mark> ow)	
		Indica	ators (A)		vs		Indica	ators (B)		
	% of budg for enviro	get of local	governmen	t allocated	νs	R&D exp	enditure in	% of GDP		
1			□5	Пз			5	D 7	9	
	% of budg	get of local	governmen	t allocated	Vs	Use of el	ectricity pe	r GDP		
2			□5	□3	\square_1				9	
	% of budg	get of local	governmen	t allocated	νs	Use of w	ater per GD	P		
3	9			□3					9	
	% of budg for enviro	get of local onment	governmen	t allocated	٧s	Gross cit	y product p	er capita		ĺ
ı	e		□5	□3					9	
	% of budg for enviro	get of local onment	governmen	t allocated	γs	Househo	ld below po	overty line		
5	e			3						
	R&D expe	enditure in	% of GDP		γs	Use of el	ectricity pe	r GDP		
5	9		□5	□3	\Box_1				9	
_	R&D expe	enditure in	% of GDP		Vs	Use of w	ater per GD	P		
7	9			□3	\Box_1				9	
	R&D expe	enditure in	% of GDP	•	γs	Gross cit	y product p	er capita		
5	9		□5	□3	\Box_1				9	
	R&D expe	enditure in	% of GDP		Vs	Househo	old below po	overty line		
,	9		□5	□3					9	
	Use of ele	ectricity pe	r GDP		Vs	Use of w	ater per GD	P		
*	9		5	Пз			5		9	
	Use of ele	ectricity pe	r GDP		Vs	Gross cit	y product p	er capita		
+	9		5	3			5		9	
	Use of ele	ectricity pe	r GDP		Vs	Househo	ld below po	overty line		
*	9		5	3			5		9	
#	Gross city	product p	er capita		Уs	Gross cit	y product p	er capita		
"	9		5	□3					9	
#	Gross city	product p	er capita		γ_s	Househo	ld below po	overty line		
.*	9				\square_1					
#	Househol	d below po	overty line		νs	Househo	ld below po	overty line		
	9			□3	\Box_1					
			flacer f			nte if if				T

				Your	nputs	Below					
	Which ir ii	ndicator d n the aspe	o you thir ect of " <mark>Pe</mark>	ık is <mark>more</mark> ople & Url	Importa ban Livi	nt for me ng"? (Cho	asuring a	City's S at each ro	<mark>martness</mark> ow)		
		Indica	itors (A)		vs		Indica	tors (B)			
	Number o	of doctors p	er 1000 po	pulation	Vs	Number	Number of hospitals per 1000 population				
-	9		5	□3	\Box_1		5		9		
	Number o	of doctors p	er 1000 po	pulation	Vs	Number o governm	of environm ent per	nental staff	s in city		
-	9		□s	Π3	\Box_1	\square_3	5		D 9		
	Number o	Number of doctors per 1000 population				% of indu control re	istries comp egulations	lied with e	emission		
•	9	□7	□5	3		□3	5	□7	е		
	Number o	Number of doctors per 1000 population				% of veh control re	icles compli egulations	ed with er	nission		
4	9		5	□3		□3	5		_ 9		
	Number o	of doctors p	er 1000 po	pulation	Vs	Adult lite	eracy rate				
3	9		□5	□3	\Box_1	□3	5		D 9		
	Number o	Number of hospitals per 1000 population				Number of environmental staffs in city government per					
0	9		□5	□3	\square_1	□3	□5		D 9		
7	Number of hospitals per 1000 population			Vs	% of indu control re	istries comp egulations	lied with e	emission			
	9		□5	3		□3	5		е		
	Number of hospitals per 1000 population			Vs	% of veh control re	icles compli egulations	ed with er	nission			
8	9		5				5		D 9		
	Number o	Number of hospitals per 1000 population				Adult lite	eracy rate				
,	9			□3	\square_1		5		еП		
	Number of environmental staffs in city government per				Vs	% of industries complied with emission control regulations					
U	9	□7	5	З		□3	5	□7	9		
	Number o governme	of environm ent per	nental staff	s in city	Vs	% of vehicles complied with emission control regulations					
1	- 9		5	□3			5		9		
	Number o governme	of environm ent per	nental staff	s in city	γs	Adult lite	eracy rate				
2		\square_7	□5	□3			5		D 9		
-	% of indu control re	stries comp gulations	olied with e	mission	Vs	% of veh control re	icles compli egulations	ed with er	nission		
5	9		□5	□3	\Box_1	□3			D 9		
	% of indu control re	stries comp gulations	lied with e	mission	Vs	Adult lite	eracy rate				
4	9		5	□3			5		9		
_	% of vehi control re	cles compli gulations	ied with en	nission	γs	Adult lite	eracy rate				
5	9		□5	□3					D 9		

Please Leave your comments, if there is any

42

1				You	r Imputs	Below				Retur Cover
	Whie Smar	ch indica <mark>tness in</mark>	tor do yo the asp	ou think is ect of "I	s More Im nfrastruc	portant ture"? (for meas Choose o	suring a	City's ch row)	
		Indica	ators (A)		Compare A & B		Indica	ators (B)]
	Energy co buildings	nsumption	n of residen	itial	νs	Energy-e	fficient bui	lding stand	ards]
1	e	□_7							□ ₉	
_	Energy co buildings	nsumption	n of residen	itial	Ъs	Green sp	aces per ca	pita		1
2	□,		□₅				□ ₅			
	Energy co buildings	nsumption	n of resider	itial	Vs	Accessib	ility of sma	rt grid		1
3							□ ₅		□ ₉	-
_	Energy-ef	ficient buil	ding stand	ards	γs	Green sp	aces per ca	pita		1
4		□_7						□_7		
_	Energy-ef	ficient buil	ding stand	ards	Vs	Accessib	ility of sma	rt grid		1
5								□_7	$\Box_{\mathfrak{s}}$	
_	Green spa	ices per ca	pita		γs	Accessib	ility of sma	rt grid		1
6	□ ₉								□ ₉	

				Your	Imputs	Below				Return to Coverpage
	Wh Smart	nich indica <mark>ness in t</mark>	ator do yo : <mark>he aspe</mark> o	ou think is ct of "Ene	More Im rgy & Mo	portant obility	for meas ? (Choose	uring a C one at e	<mark>City's</mark> ach row)	
		Indica	ators (A)		Compare A & B		Indica	ators (B)		
	Share of r use	enewable	energy in to	otal energy	γs	CO2 per	capita from	energy use	2	
1								□_7		
	Share of r use	enewable	energy in to	otal energy	γs	Green m	obility share	2		
2										
	Share of r use	enewable	energy in to	otal energy	γs	E-vehicle	e in commer	cial vehicle	shares	
3			\square_{5}		\Box_1					
	CO2 per c	apita from	energy use	•	γ_s	Green m	obility share	9		
4	□,								9	
	CO2 per c	apita from	energy use	•	γ_{s}	E-vehicle	e in commer	cial vehicle	shares	
5									e	
6	Green mo	bility share	2		Vs	E-vehicle	e in commer	cial vehicle	shares	
									₉	
L										1

			Your	Inputs	Below					
	Which indica Smartness	ator do yo in the as _l	ou think is pect of " <mark>E</mark>	more lı nvironn	nportant nent"? (C	for meas hoose or	suring a ne at eac	City's h row)		
	Indica	ators (A)		vs		Indica	ators (B)			
	Air quality (indivate Suspended Particles	d by SO2, T s etc.)	otal	⊐ ∕vs	% of pop and clear	ulation wit า water	h access to	adequate		
1		5	Пз		□3	5		9		
,	Air quality (indivated by SO2, Total Suspended Particles etc.)		Vs	Water qu contents	Water quality (measured by BOD, COD contents etc.)					
2	9 7	5	3		3	5		9		
	Air quality (indivated by SO2, Total Suspended Particles etc.)			γ_s	Per capit	al waste ge	eneration			
3		5	3			5		9		
	Air quality (indivate Suspended Particles	d by SO2, T etc.)	otal	γ_s	% of tota	l municipal	l solid wast	e (MSW)		
4		5 5	Пз				D 7	9		
	Air quality (indivate Suspended Particles	d by SO2, T etc.)	otal	Vs	% of tota	l waste wa	ter treated	or recycled		
5	\square_9 \square_7	□5	□3			□5		9		
	% of population wit and clean water	Vs	Water qu	Water quality (measured by BOD, COD						
6		□5	3			□5	D 7	9		
	% of population with access to adequate and clean water		γ_s	Per capit	al waste ge	eneration				
7			□3					\square_9		
	% of population wit and clean water	νs	% of tota collected	l municipal & treated	l solid wast	e (MSW)				
\$	$\square_9 \qquad \square_7$		3			□5		9		
	% of population with access to adequate and clean water		Vs	% of tota	% of total waste water treated or recycled					
9		5	□3		3			9		
	Water quality (meas	sured by BC	DD, COD	Vs	Per capit	al waste ge	eneration			
0		5	Π3							
	Water quality (meas	sured by BC	DD, COD	Vs	% of tota	% of total municipal solid waste (MSW)				
11		5						Пэ		
	Water quality (meas	sured by BC	DD, COD	Vs	% of tota	l waste wa	ter treated	or recycled		
2		5						Пэ		
	Per capital waste ge	eneration		Vs	% of tota	% of total municipal solid waste (MSW)				
13			□3					9		
	Per capital waste ge	eneration		Vs	% of tota	l waste wa	ter treated	or recycled		
14	□ ₉ □ ₇	5	Π3		Π3	5		9		
	% of total municipal	− ∕vs	% of tota	l waste wa	ter treated	or recycled				
	collected & treated									